Author's response to reviews

Title: Fitness consultations in routine care of patients with type 2 diabetes in general practice: An 18-month non-randomised intervention study

Authors:

 Henning Lohmann (Lohmann@dadlnet.dk)
 Niels F Olivarius (olivarius@sund.ku.dk)
 Volkert Siersma (siersma@sund.ku.dk)

Version: 5 Date: 1 November 2010

Author's response to reviews:

"Major revisions (we require the author to make these changes)
Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)"

I have changed Niels de Fine Olivarius to Niels F Olivarius in Manuscript and submission system.

"Authors' contributions: Please indicate that all authors read and approved the final manuscript."

I have added the statement in the end of the section "Authors Contribution"

After doing a last proof I found this error in the percent numbers in the manuscript:

"There was a more substantial increase of 8.3% in the subgroup (n=53) without function-limiting pain and 6.3 % in the subgroup (n=61) without CVD (Table 2)."

Has been changed in the final manuscript to:

"There was a more substantial increase of 7.9% in the subgroup (n=53) without function-limiting pain and 5.2 % in the subgroup (n=61) without CVD (Table 2)

I will be happy if you do the Minor revisions."